Author’s response to reviews

Title: Use of Natural language processing to improve predictive models for imaging utilization in children presenting to the Emergency Department

Authors:

Xingyu Zhang (zhangxyu@umich.edu)
Fernanda Bellolio (Bellolio.Fernanda@mayo.edu)
Pau Medrano-Gracia (p.medrano@auckland.ac.nz)
Konrad Werys (konrad.werys@cardiov.ox.ac.uk)
Sheng Yang (yasheng@umich.edu)
Prashant Mahajan (pmahajan@med.umich.edu)

Version: 1 Date: 22 May 2019

Author’s response to reviews:

Dear Editor ,

We have added the elements needed in the declaration section including-

Consent for publication: Not applicable

Ethical approval and consent to participate:

This study was a secondary analysis of a public database and did not require ethical approval.

Availability of data and materials:

The NHAMCS-ED dataset can be accessed through the website of the US Centers for Disease Control and Prevention (CDC) (https://www.cdc.gov/nchs/ahcd/index.htm).

A new version has been uploaded.

Thanks

Xingyu Zhang and Co-authors